Author's response to reviews

Title: The burden of disease profile of residents of Nairobi's slums: Results from a Demographic Surveillance System

Authors:

Catherine Kyobutungi (ckyobutungi@aphrc.org)
Abdhalah Kasiira Ziraba (aziraba@aphrc.org)
Alex Ezeh (aezeh@aphrc.org)
Yazoume Ye (yyazoume@aphrc.org)

Version: 6 Date: 7 March 2008

Author's response to reviews:

Formatting changes requested:
1) Reference citing in the text: Reference No. 20 has been changed and it is now No 31. All other references thereafter have been readjusted accordingly in the text and the bibliography
2) Authors’ contributions - Please indicate that all authors read and approved the final manuscript: Done!
3) Reference list: "el al" has been removed from Reference No 34 and the full list of authors has been provided
4) Figures cropping: Figures 1 and 2 have been cropped to fill the page size as suggested